Author’s response to reviews

Title: A water extract of Samchulkunbi-tang attenuates airway inflammation by inhibiting iNOS and MMP-9 activities in an ovalbumin-induced murine asthma model

Authors:

Mee-Young Lee (cozy37@gmail.com)
In-Sik Shin (dymshin@kiom.re.kr)
Hye-Sun Lim (qp1015@kiom.re.kr)
Hyeun-Kyoo Shin (8282tongil@daum.net)

Version: 2 Date: 21 September 2012

Author’s response to reviews: see over
Research article

MS: 1207780713809756

Title: A water extract of Samchulkunbi-tang attenuates airway inflammation by inhibiting iNOS and MMP-9 activities in an ovalbumin-induced murine asthma model

BMC Complementary and Alternative Medicine (Section: Basic research)

Dear Editors

Thank you very much for your editorial decision letter, which also included the reviews of our manuscript by referees. We have made the changes as suggested by the reviewers. The changes are marked in blue in the revised text. We have made during the revision in a point-by-point response to each of the comments.

We hope the revisions made the responses provided are satisfactory, and our manuscript is now acceptable for publication in the BMC Complementary and Alternative Medicine.

Please, let us know if further revisions are needed.

Once again, thank you for all your help. We look forward to hearing from you.

Sincerely yours,

cozy37@gmail.com
Editorial comments

Editorial requests:

Thank you for including an ethics statement in your manuscript. However, could you please confirm that specific ethical approval was granted for the protocols used in your study, and amend your ethics statement to reflect this. Please also include the name of the ethics committee that approved your study in this statement.

→ We appreciate your comment. As commented by editor, we added this statement in Materials and Methods (page 04, line 20-24)